Author's response to reviews

Title: Ice-Pick headache syndrome, an 'indomethacin-responsive' headache syndrome which responded to a less toxic NSAID, implications for clinical practice: A case report.

Authors:

Mortimer B O'Connor (mortimeroconnor@gmail.com)
Elizabeth Murphy (mortimer_o_connor@hotmail.com)
Mark J Phelan (meganbreen01@hotmail.com)
Michael J Regan (Regan.Michael@sivuh.ie)

Version: 2 Date: 25 June 2007

Author's response to reviews:

1). restructuring the title so it ends in a 'a case report.'
2). declare in the Acknowledgements section that informed written consent was received from the patient for publication of the manuscript